Author’s response to reviews

Title: HIV Positive Sero-Status disclosure and its determinants among People Living with HIV/AIDS following ART Clinic in Jimma University Specialized Hospital, Southwest Ethiopia: A facility-based cross-sectional study

Authors:
Tamiru Tesfaye (tamirutesfaye00@gmail.com)
Jiregna Darega (jiregnadarega@yahoo.com)
Tefera Belachew (terabelachew@gmail.com)
Abebe Abera (abeef2011@gmail.com)

Version: 2 Date: 15 Sep 2017

Author’s response to reviews:

GENERAL COMMENTS:

Comments 1: “I have gone through the revised manuscript, authors have responded to the comments I raised and addressed them adequately; however the language has not been revised. This manuscript should be sent to someone to edit it if it is to be published”.

Answers from authors: Thank you for your comments and concerns. As previously, we responded English language is not our native language, we faced difficulties all the time on grammars, punctuations, etc. Again, in line your second time comments, we used other two persons that are English language experts or teachers that teaching at university level. We totally accept the comment and we tried to use available chance we have to edit it. Thank you again for your concern!

Generally, thank you for your best comments!